Author’s response to reviews

Title: Concomitant glenohumeral injuries in Neer type II distal clavicle fractures

Authors:

Tobias Helfen (tobias.helfen@med.uni-muenchen.de)

Georg Siebenbürger (georg.siebenbuerger@med.uni-muenchen.de)

Florian Haasters (florian.haasters@med.uni-muenchen.de)

Wolfgang Böcker (wolfgang.boecker@med.uni-muenchen.de)

Ben Ockert (ben.ockert@med.uni-muenchen.de)

Version: 3 Date: 12 Jan 2018

Author’s response to reviews:

Dear Editorial Office,

1. We included title page

2. Discussion and Results has been separated

3. Abbreviation list completed

4. Full name of ethics committee added, informed consent added

5. Consent for publication changed

6. Authors contributions more detailed

7. Figures separated and legends repositioned as well as figures cited in manuscript

Sincerely

T. Helfen